MCC Operations Status Meeting
Friday, 25-Jan-2008

This afternoon at the  4:15pm meeting:

Everyone: 

Paul Emma has presented the controls software group with a challenge for next week. By next Friday,  February 1st, he would like to see  the LCLHOME Screen  that displays no alarms, “green”!  If we are successful there will be a celebration.

BPMS:  Steve Smith and Bob Trailer
1. Replaced LI28 701,801,901 CAMAC modules.

2. Next, will replace LI29 CAMAC modules for bpms
Laser: Sheng Peng and Zen Szalata

The users needs to lower the peak to peak pixel in order to get rid of the alarm state.

LLRF: Mike Stanek,  Joe Frisch and Ed Miller, Ron Akre

1. Joe Frisch will write a temporary matlab program that reads an SLC database to get the driver line phase and will use this to set the rf reference phase in his matlab program. The long term solution is still to have Ed Miller modify the compensation slow feedback.

2. Next Wednesday RF 24-1,2,3 will be timed in. Only PACs without feedback loops locally. The old feedback looks however, will still exist. 

3. TCAV3 needs to be timed-in.

MPS: Arturo Alarcon and Mike Cecere
1. Toroid comparator is working in bc1.

SCORE: Debbie Rogind

Debbie would like to test and possibly release a new version of SCORE to cleanup the comment field from intermittently being overwritten with the newly saved Comment field.

Wire Scanners: Arturo Alarcon

1. WS01 calibration timeout. Arturo knows about this, and will look at  the calibration. However, the wire scanner will work fine without calibration.

2. The LI28 wire scanners are working

3. Wire scanner/Aida RMATRIX, A to B for bpm 13 in the injector has been fixed. Rick Iverson did a dbedit for a wire scanner to solve this problem. 
This morning at the 8:00am meeting:
PEPII
BIC: Lazmo

1.  Attenuation problem occurred yesterday and was fixed by Mike.
2.  BIC was having trouble receiving the waveform data from the Bunch Current Monitor even when the data on and EPICS display from both the HER and LER Bunch Current Monitor looked good. 

3.  BIC PB60:HEARTBEAT was alarming.  It appears that the process on the BCM was hung up as this pv was changing state from alarm to warning to no alarm. A reboot of the BCM cleared up the problem temporarily.

Note: A similar problem occurred in December 2007. At that time the problem was related to the  BIC ioc running out of network buffers.
Computer Systems: Jingchen  Zhou
1. Luminosity Web display is down. Jingchen is looking into this problem.
DbInstall – Ken Underwood

1.  Fallout related to bunch timing which jumped 100 ps  This is not clearly understood. Collision timing slope shows problem. Bob Trailer and Mike Browne are looking at this problem.

LCLS
Laser: Sheng Peng and Zen Szalata

1. When an alarm (red) was displayed on the lcls home page in the IN20 laser gun region, the user could not find where the alarm originated by going to the Laser IN20 top level display. I showed the users how to find this by using the  Alarm Handler to drill down to pv with the alarm set was related to a Dark image pv. This pv is located on the Laser IN20 VCC panel, and show as an alarm in red. However, the operators would like alarm to be propagated from the lower level VCC display up to the top level IN20 Laser display. This would allow the alarm is quickly understood.

LLRF:  Mike Stanek, Ron Akre and Ed Miller

1. The problem of the phase shifting is now understood more clearly. The fix is a modification to the compensation slow feedback in the old control system to set the NULL pointer at sector 20. There were a few solutions knocked around including hardware. However, the hardware solutions were difficult and none were very good. 

2. There was a sub-booster 22 problem with the phase reading. This was corrected by moving the sub-booster 90deg and then moving the phase 90deg in the opposite direction. This fix has been reported before and a complaint about how clunky this user interface, of setting PDES on the old VMS system. PaulE would like to solve this problem by moving as quickly as possible away from the reliance on old control system. 

Mike Stanek will contact Ed Miller regarding the compensation slow feedback.

SCORE: Debbie Rogind

When configuration files are saved, it appears that the file is being saved two or three times with the same names, but at different times, even when this did not occur. Another symptom noticed was that a file saved at 4pm with a text description of “BXG…”, comes up on the SCORE display as a file saved at around 3pm that same day. So the names of the files seemed to be mixed up. Debbie looked at the problem and discovered that the problem began on Jan 18th, and so the configuration files saved since then are unusable. The files since Jan 18th are being removed.
Torids:  Mike Cecere  and Arturo Alarcon

1. When there is no beam present, the toroids read charge that is approximately 10% of normal beam. PaulE said that he would be happy to give time to Mike for doing correlations between the modulator and the toroids or other studies that Mike may want to do.  

Wire Scanners: Arturo Alarcon and Nancy Spencer

1. In December 2007 a dbedit was done on SLC to modify an LI27 644 Wire Scanner field. This change was never put into the SLC database permanently and so after the dbinstall yesterday a BPM in LI27 reported an error related to wire scanner 644. A dbedit was done to fix this problem, but it is unknown if this has been put into the database permanently.
